Author's response to reviews

Title: Bronchial obstruction secondary to idiopathic scoliosis in a child.

Authors:

Dr. Saad Alotaibi (saalotai@hotmail.com)
Dr. James Harder (james.harder@calgaryhealthregion.ca)
Dr. Sheldon Spier (Sheldon.Spier@CalgaryHealthRegion.ca)

Version: 8 Date: 11 April 2008

Author's response to reviews:

Cover letter

1. The flow volume curve figure was edited to include an arrow indicating the site of the scoping and concavity in the expiratory part of the curve as asked by the reviewer.

2. In the discussion part the following statement was corrected to "If there is obstruction, direct visualization through a flexible bronchoscope would help identify the precise site and severity of the airway obstruction and guide further management such as spinal orthosis or insertion of spinal rods. This was advised by the reviewer in his comment."

Thanks
Your's
Saad Alotaibi.